Author's response to reviews

Title: The effects of anterior vacuum disc on surgical outcomes of degenerative versus spondylolytic spondylolisthesis: at a minimum two-year follow-up

Authors:

Tung-Yi Lin (ross_1222@hotmail.com)  
Jen-Chung Liao (jcl1265@adm.cgmh.org.tw)  
Tsung-Ting Tsai (tsai1129@gmail.com)  
Meng-Ling Lu (mengling@adm.cgmh.org.tw)  
Chi-Chien Niu (niuchien@adm.cgmh.org.tw)  
Wen-Jer Chen (chenwenj@adm.cgmh.org.tw)  
Lih-Hui Chen (lhchen2132@adm.cgmh.org.tw)

Version: 6  
Date: 17 September 2014

Author's response to reviews: see over
Dear Editor:

Thank you for the comments concerning our manuscript entitled "The effects of anterior vacuum disc on surgical outcomes of degenerative versus spondylolytic spondylolisthesis: at a minimum two-year follow-up" (MS: 5405397251170378). We revised manuscript conforms to the journal style to improve the quality of our article and the completed STROBE checklist as an additional file had been upload. If there is any question, please do not hesitate to let me know. Thank you very much!

The manuscript conforms had changed as below:

Abstract section:
We added ":" after the Background, Methods, Results, Conclusion and Key words

Methods:
The previous "material and methods" had been changed to Methods

Results:
Table I, Table II and Table III changed to Table 1, Table 2 and Table 3

Reference section:
The style had been changed to BMC form.
"Ethic statement" had been deleted

Sincerely yours,

Sep 17, 2014

____________________
Tung-Yi Lin, Jen-Chung Liao, Meng-Ling Lu, Chi-Chien Niu, Lih-Hui Chen, Wen-Jer Chen

Department of Orthopedic Surgery, Chang Gung Memorial Hospital, Chang Gung University

No._5, Fu-Shin Street

Kweishian, Taoyuan 333, Taiwan

TEL: 886-3-3281200, ext 2423; FAX: 886-3-3278113;

E-Mail to Dr. Jen-Chung Liao: jcl1265@adm.cgmh.org.tw